Author's response to reviews

Title: Prevalence, correlates and pattern of Hepatitis B among antenatal clinic attenders in Yaounde-Cameroon.

Authors:

Nelson J Fomulu (ikomifred@gmail.com)
Frederick L.I Morfaw (ikomi_fred@yahoo.com)
Judith N Torimiro (itorimiro@yahoo.co.uk)
Philip N Nana (njotangnanaphilip@yahoo.com)
Mve V. Koh (mve_koh@yahoo.com)
William Takang (takangwilliam1@yahoo.com)

Version: 2 Date: 26 December 2012

Author's response to reviews:

Dear sir,

In accordance to your request, i am providing a separate email address for each of the corresponding authors.

I hope you will now find our article suitable for consideration for publication

Sincerely,

Frederick Morfaw